Author's response to reviews

Title: Season of birth is different in Inuit suicide victims born into Traditional than into Modern Lifestyle. A register study from Greenland.

Authors: Karin S Björkstén (Karin.Sparring.Bjorksten@ki.se)  
Peter Bjerregaard (pb@si-folkesundhed.Dk)

Version: 5  
Date: 14 April 2015

Author's response to reviews:

Dear Dr Wouter van Ballegooijen  
Thank you for your nice words.

“Line 232-234  
I got confused by these sentences, because it can be interpreted as if you are statistically comparing the months March-June with the other months.”

Line 232-234 reads  
“The average of this suicide rate for the nadir months March-June was calculated and compared with the average of the remaining eight months for the Traditional Lifestyle cohort and for the Modern Lifestyle cohort.”

New version  
“The average of this suicide rate for the nadir months March-June and for the remaining eight months July-February was calculated.”

Karin Sparring Björkstén

Karin Sparring Björkstén